Washington University School of Medicine Digital Commons@Becker

Open Access Publications

2014

Age, gender, and cancer but not neurodegenerative and cardiovascular diseases strongly modulate systemic effect of the Apolipoprotein E4 allele on lifespan

Alexander M. Kulminski Duke University

Konstantin G. Arbeev Duke University

Irina Culminskaya Duke University

Liubov Arbeeva Duke University

Svetlana V. Ukraintseva Duke University

See next page for additional authors

Follow this and additional works at: http://digitalcommons.wustl.edu/open_access_pubs

Recommended Citation

Kulminski, Alexander M.; Arbeev, Konstantin G.; Culminskaya, Irina; Arbeeva, Liubov; Ukraintseva, Svetlana V.; Stallard, Eric; Christensen, Kaare; Schupf, Nicole; Province, Michael A.; and Yashin, Anatoli I., ,"Age, gender, and cancer but not neurodegenerative and cardiovascular diseases strongly modulate systemic effect of the Apolipoprotein E4 allele on lifespan." PLoS Genetics.10,1. e1004141. (2014).

http://digitalcommons.wustl.edu/open_access_pubs/2181

This Open Access Publication is brought to you for free and open access by Digital Commons@Becker. It has been accepted for inclusion in Open Access Publications by an authorized administrator of Digital Commons@Becker. For more information, please contact engeszer@wustl.edu.

Authors

Alexander M. Kulminski, Konstantin G. Arbeev, Irina Culminskaya, Liubov Arbeeva, Svetlana V. Ukraintseva, Eric Stallard, Kaare Christensen, Nicole Schupf, Michael A. Province, and Anatoli I. Yashin

Age, Gender, and Cancer but Not Neurodegenerative and Cardiovascular Diseases Strongly Modulate Systemic Effect of the Apolipoprotein E4 Allele on Lifespan

Alexander M. Kulminski^{1,2,3}*, Konstantin G. Arbeev^{1,3}, Irina Culminskaya^{1,3}, Liubov Arbeeva¹, Svetlana V. Ukraintseva^{1,2,3,4}, Eric Stallard^{1,3}, Kaare Christensen^{5,6}, Nicole Schupf⁷, Michael A Province⁸, Anatoli I. Yashin^{1,2,3,4}

1 Center for Population Health and Aging, Duke University, Durham, North Carolina, United States of America, **2** Institute for Genome Science and Policy, Duke University, Durham, North Carolina, United States of America, **3** Social Science Research Institute, Duke University, Durham, North Carolina, United States of America, **4** Duke Cancer Institute, Duke University, Durham, North Carolina, United States of America, **5** The Danish Aging Research Center, University of Southern Denmark, Odense, Denmark, **6** Department of Clinical Genetics and Department of Clinical Biochemistry and Pharmacology, Odense University Hospital, Odense, Denmark, **7** Taub Institute for Research on Alzheimer's Disease and the Aging Brain, Columbia University Medical Center, New York, New York, United States of America, **8** Washington University School of Medicine, Division of Statistical Genomics, St. Louis, Missouri, United States of America

Abstract

Enduring interest in the Apolipoprotein E (ApoE) polymorphism is ensured by its evolutionary-driven uniqueness in humans and its prominent role in geriatrics and gerontology. We use large samples of longitudinally followed populations from the Framingham Heart Study (FHS) original and offspring cohorts and the Long Life Family Study (LLFS) to investigate genderspecific effects of the ApoE4 allele on human survival in a wide range of ages from midlife to extreme old ages, and the sensitivity of these effects to cardiovascular disease (CVD), cancer, and neurodegenerative disorders (ND). The analyses show that women's lifespan is more sensitive to the e4 allele than men's in all these populations. A highly significant adverse effect of the e4 allele is limited to women with moderate lifespan of about 70 to 95 years in two FHS cohorts and the LLFS with relative risk of death RR = 1.48 ($p = 3.6 \times 10^{-6}$) in the FHS cohorts. Major human diseases including CVD, ND, and cancer, whose risks can be sensitive to the e4 allele, do not mediate the association of this allele with lifespan in large FHS samples. Non-skin cancer non-additively increases mortality of the FHS women with moderate lifespans increasing the risks of death of the e4 carriers with cancer two-fold compared to the non-e4 carriers, i.e., RR = 2.07 (p = 5.0×10^{-7}). The results suggest a pivotal role of non-sex-specific cancer as a nonlinear modulator of survival in this sample that increases the risk of death of the ApoE4 carriers by 150% ($p = 5.3 \times 10^{-8}$) compared to the non-carriers. This risk explains the 4.2 year shorter life expectancy of the e4 carriers compared to the non-carriers in this sample. The analyses suggest the existence of age- and gender-sensitive systemic mechanisms linking the e4 allele to lifespan which can non-additively interfere with cancerrelated mechanisms.

Citation: Kulminski AM, Arbeev KG, Culminskaya I, Arbeeva L, Ukraintseva SV, et al. (2014) Age, Gender, and Cancer but Not Neurodegenerative and Cardiovascular Diseases Strongly Modulate Systemic Effect of the Apolipoprotein E4 Allele on Lifespan. PLoS Genet 10(1): e1004141. doi:10.1371/journal.pgen.1004141

Editor: Nicholas J. Schork, University of California San Diego and The Scripps Research Institute, United States of America

Received September 3, 2013; Accepted December 6, 2013; Published January 30, 2014

Copyright: © 2014 Kulminski et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Funding: The research reported in this paper was supported by Awards Number R01AG030612 and 2U01AG023712 from the National Institute on Aging. The Framingham Heart Study (FHS) is conducted and supported by the National Heart, Lung, and Blood Institute (NHLBI) in collaboration with the FHS Investigators. This manuscript was prepared using a limited access FHS data obtained from the NHLBI through dbGaP. This manuscript was not prepared in collaboration with investigators of the FHS and does not necessarily reflect the opinions or views of the FHS, Boston University, or the NHLBI. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

* E-mail: Alexander.Kulminski@duke.edu

Introduction

The Apolipoprotein E (ApoE) common polymorphism (e2, e3, and e4) is one of the most studied genetic variants in humans. The interest in this polymorphism is two-fold. First, the functional diversity of the ApoE polymorphism appears to be a unique signature of humans with no coding variation in this gene even in human's closest ancestries in which the monomorphic ApoE sequence resembled human's e4 allele [1,2]. Understanding the functional diversity of the ApoE gene, thus, can help in gaining insights on human evolution. Second, the ApoE polymorphism is of fundamental interest for geriatrics and gerontology because of

its profound role in human diseases in late (post-reproductive) life and lifespan.

Most consistent associations were reported for the detrimental effect of the e4 allele on Alzheimer disease [3–5]. Studies also mostly documented a detrimental role of the e4 allele in cardiovascular health [6,7] although a protective role of this allele was also reported [6,8]. The e4 allele was associated with human lifespan and longevity in a number of studies [9–19]; some studies reported, however, no significant effect [20–22] (see also http:// genomics.senescence.info/longevity). Studies of the role of the e4 allele in human longevity were mostly limited to comparing frequencies of genotypes in long-living individuals and younger

Author Summary

Discovering genetic origins of healthspan and lifespan could lead to breakthroughs in increasing the years of healthy and long life. In this paper we characterize the association of the e4 allele of the well-studied ApoE gene with lifespan in two generations of participants of large longitudinal studies, the Framingham Heart Study and the Long Life Family Study, and investigate the role of major human diseases such as cardiovascular disease, cancer, and neurodegenerative disorders in this association. This wide range of systemic analyses is possible given the large sample with directly genotyped ApoE polymorphism available from these studies (N = 9841, with 2557 deaths). The analyses show that women's lifespan is more sensitive to the e4 allele than men's in these populations. However, the strongly adverse effect of the e4 allele is not observed for all women, but only for those 70 to 95 years old. Cardiovascular disease, cancer, and neurodegenerative disorders do not mediate the association of the e4 allele with lifespan. However, cancer, but not cardiovascular and neurodegenerative diseases, non-additively enhances this effect resulting in 4.2 years of difference in mean lifespan for the e4 allele carriers compared to the non-carriers.

controls [23], a strategy which has limitations [24]. Studies examining survival of older individuals carrying the e4 allele are rare (notably, [17,18]). Sexual dimorphism of the ApoE gene in human survival has not been widely studied so far (see [17] and references therein).

Since the e4 allele may be involved in regulation of such common diseases in the elderly as dementia and cardiovascular diseases (CVD), it is often assumed that the detrimental effect of the e4 allele on human longevity is mediated by these diseases (e.g., [11,14,25]). Studies of the systemic effect of the e4 allele and major human diseases on lifespan in the same samples are rare [16,17] primarily because they require large samples of genotyped individuals followed for a long period of time to have sufficient number of events.

Despite the detrimental role of the e4 allele in human health and longevity, this allele continues to be widespread in human population [26]. The persistence of this allele has been proposed to be a result of balancing selection implying that the e4 allele should be also evolutionarily advantageous with a beneficial role in early life [27–30].

In this work we examine three inter-related problems which, taken together, address the systemic role of the e4 allele in human lifespan. First, we investigate gender-specific effects of the ApoE4 allele on survival in a wide range of ages starting from midlife to extreme old ages. Second, we examine whether major human diseases such as CVD, cancer, and neurodegenerative disorders (ND) can explain (i.e., mediate) the effect of the e4 allele on survival. Third, we investigate whether these diseases can modulate the e4-specific survival non-additively. This wide range of systemic analyses is possible given the large sample with directly genotyped ApoE polymorphism available for the analyses and selected from the Framingham Heart Study (N = 5182) and the Long Life Family Study (N = 4659) followed longitudinally for up to 60 years with a total of 2557 deaths.

Results

given in Table 1. Table 1 shows that the proportion of the e4 allele carriers is about the same regardless of gender in FHS and FHSO cohorts at the time of biospecimens collection, i.e., 20.6% FHS men, 22.3% FHSO men, 22.8% FHS women, and 22.7% FHSO women carry the e4 allele. The FHSO sample of genotyped survivors (mean age of about 50 years) was, however, about 20 years younger than that in the FHS at the time of biospecimens collection (Table 1) indicating no strong e4-specific survival selection by that time in the FHS and FHSO survivors. The proportion of the e4 allele carriers in the LLFS was the largest in spouses; it was (significantly [31]) smaller in children of the long-living individuals compared to younger population of long-living individuals compared to younger populations.

Empirical Age Patterns of Survival of the FHS and LLFS Men and Women

Our empirical analysis showed no consistent detrimental effect of the e4 allele across ages on the survival of men either in the FHS or FHSO cohorts (Figures 1A and 1C). Contrary to men, the e4 female carriers have shorter lives than the non-carriers (Figures 1B and 1D). An important result is that the role of the e4 allele in survival can change with age. Specifically, there is no e4-specific difference in survival of either the FHS men or women at ages 95 years and older (Figures 1A and 1B). The e4 allele does not affect survival at ages 70 years and younger (Figure 1D) either.

Analysis of survival age patterns of the LLFS male and female offspring/spouses directly supports these observations. Specifically, the LLFS female offspring and spouses carrying the e4 allele show worse survival than those who do not carry this allele (Figure 2D). Survival of the LLFS male offspring and spouses is not sensitive to this allele (Figure 2C).

To better understand survival age patterns of the LLFS participants from the parental generation (Figures 2A-B), one should keep in mind that this is a population selected for its exceptional chances to live a long life based on family history and their own survival to old ages (see Methods). Accordingly, this population resembles the subpopulation of individuals who survive to the very old ages in the FHS original cohort rather than the entire sample of a normal population in this cohort. Then, an important result is that the LLFS women selected for their chances of exceptional longevity (Figure 2B) and the long-living women in the FHS original cohort (represented in Figure 1B by a tail of the survival age pattern) have the same lifespan regardless of whether they carry the e4 allele. When analyzing survival age patterns one should also consider the possibility of survival selection in aging cohorts; if this selection is sensitive to a specific genetic variant then we may have biased empirical age patterns for carriers of genotypes from this variant particularly at advanced ages. Then, although the lifespans of the long-living LLFS men may be sensitive to the e4 allele (Figure 2A), further analyses are necessary (see next subsection) to determine whether this effect is real.

Thus, Figures 1B and 2B document an important result that survival of long-living women participating in the FHS (see upper tail in Figure 1B) and LLFS is insensitive to the e4 allele. Figures 1D and 2D show another remarkable result that the effect of the e4 allele on survival in the FHSO and LLFS offspring/ spouses is pronounced: (i) starting at the same age, 70 years and (ii) in women only.

Risks of Death of the FHS and LLFS Men and Women

We evaluated the sensitivity of the survival of the long-living LLFS men to the e4 allele seen in Figure 2A. Evaluation of the relative risk (RR) of death for the e4 allele carriers using a model

The proportions of the ApoE4 allele carriers (see Methods) and the allele-specific proportions of deaths, CVD, cancer, and ND are **Table 1.** Proportions of the ApoE4 allele carriers, mean age at the time of biospecimens collection, and the allele-specific proportions of deaths, CVD, cancer, and ND for the genotyped participants of the FHS, FHSO, and LLFS.

Study	e4	N (% [*])	Age (SD) years ^{***}	Death	Death		CVD		Cancer		ND	
				N	%	N	%	N	%	Ν	%	
Men												
FHS	no	362**	73.1 (5.4)	322	89.0	260	71.8	151	41.7	77	21.3	
	yes	94 (20.6)	74.5 (5.6)	85	90.4	66	70.2	33	35.1	26	28.3	
FHSO	no	1456	51.8 (10.2)	346	23.8	451	31.0	377	25.9	19	1.3	
	yes	418 (22.3)	52.5 (10.1)	111	26.6	148	35.4	97	23.2	17	4.2	
LLFS_P	no	567	90.0 (5.7)	276	48.7							
	yes	95 (14.4)	88.3 (5.2)	46	48.4							
LLFS_O	no	778	60.6 (8.4)	23	3.0							
	yes	203 (20.7)	60.3 (7.8)	3	1.5							
LLFS_S	no	333	65.1 (10.4)	23	6.9							
	yes	116 (25.8)	65.3 (9.7)	9	7.8							
Women												
FHS	no	619**	74.2 (5.7)	494	79.8	392	63.3	194	31.3	146	23.9	
	yes	183 (22.8)	73.8 (5.7)	155	84.7	94	51.4	48	26.2	66	36.9	
FHSO	no	1584	51.5 (10.1)	203	12.8	280	17.7	317	20.0	15	1.0	
	yes	466 (22.7)	51.1 (9.3)	81	17.4	99	21.2	86	15.5	12	2.6	
LLFS_P	no	636	91.0 (6.9)	287	45.1							
	yes	86 (11.9)	89.3 (5.7)	30	34.9							
LLFS_O	no	1069	60.4 (8.3)	23	2.2							
	yes	271 (20.2)	61.0 (7.7)	10	3.7							
LLFS_S	no	379	65.0 (13.6)	19	5.0							
	yes	126 (25.0)	64.4 (13.3)	11	8.7							

FHS = the Framingham Heart Study (FHS) original cohort; FHSO = the FHS offspring cohort.

LLFS_P = long-living parental generation of the Long Life Family Study (LLFS) participants; LLFS_O = offspring of the LLFS long-living participants; LLFS_S = spouses of the LLFS long-living participants and their offspring.

CVD = cardiovascular diseases including diseases of heart and stroke; Cancer = all sites but skin; ND = dementia and Alzheimer disease combined; SD = standard deviation.

*proportion of the ApoE4 allele carriers is in percentages;

**maximal sample size; the number of individuals with non-missing information on ND is about 1% less in the FHS and about 3% less in the FHSO.

***age at biospecimens collection at the 19th FHS, 4th FHSO and baseline LLFS examinations.

The ApoE4-allele-specific proportions of CVD, cancer, and ND are not given for the LLFS because this information was not used in this paper.

doi:10.1371/journal.pgen.1004141.t001

without adjustment for birth cohorts supported the presence of the effect (RR = 1.52, $p = 6.9 \times 10^{-3}$) in this sample (Table 2). However, adjustment for birth cohorts entirely explained this association (RR = 1.17, p = 0.319; Table 2), suggesting that this sensitivity was likely due to differential survival of the e4 carriers and non-carriers in different birth cohorts in the LLFS.

The relative risks obtained from the data in Figures 1 and 2 revealed the presence of a significant detrimental effect of the e4 allele on survival in women in the FHS (RR = 1.25, p = 0.027), FHSO (RR = 1.59, p = 2.4×10^{-4}), and LLFS offspring/spouse (LLFS_O+S; RR = 2.23, p = 5.2×10^{-3}) samples (Table 2, all). No significant effect was seen in men in either sample or in long-living women in the LLFS (Table 2, all). Pooled data from the FHS and FHSO slightly improved the significance of the estimates for women, RR = 1.36, p = 1.3×10^{-4} (Table 2, FHS+FHSO, all).

However, given the empirical evidence on the substantial role of age-related heterogeneity (Figures 1 and 2), analyses of the relative risks using the Cox proportional hazards regression model, which disregards such heterogeneity, likely underestimate the effects. A more appropriate way to address the impact of age-related heterogeneity is to consider more homogeneous groups of individuals for whom the variation of the hazards is proportional over age. Empirical evidence from independent FHS and LLFS cohorts (Figures 1 and 2) suggests selecting more homogeneous groups of individuals who died or were censored at ages: (i) younger than 95 years in the FHS (note that there were virtually no genotyped individuals with lifespans less than 70 years in this sample), (ii) 70 years and older in the FHSO and LLFS_O+S (note, virtually all genotyped participants in these samples had lifespans less than 95 years), and (iii) 70 to 95 years in the pooled sample of the FHS and FHSO.

Table 2 shows that individuals from these more homogeneous groups in each sample are at substantially larger risk of death compared to the entire sample. For example, we observe 9% increment (from RR = 1.36 to RR = 1.48) in the risk of death in the more homogeneous 70–95 year group of the FHS and FHSO women. Correspondingly, the significance of the estimate also sharply increases from $p = 1.3 \times 10^{-4}$ to 3.6×10^{-6} .

Importantly, the analyses also confirm the lack of a significant effect of the e4 allele on survival in the groups of individuals who did not belong to the selected more homogeneous groups (Table 2). Specifically, no significant effects were observed in: (a) the groups



Figure 1. Empirical age patterns of survival of the ApoE4 carriers and non-carriers in the FHS. Patterns are shown for (A and C) men and (B and D) women genotyped in (A and B) FHS and (C and D) FHSO cohorts who carry (E4) and do not carry (NoE4) the ApoE4 allele. The numbers in the insets show the total number of genotyped individuals and the number of deaths among them. doi:10.1371/journal.pgen.1004141.g001

of individuals with lifespans less than 70 years in the FHSO and LLFS_O+S, (b) individuals with exceptional survival including the entire sample of the LLFS long-living men and women (LLFS_P), and (c) individuals who were aged 95 years and older in the FHS. The lack of significant effects cannot be explained by the sample size differences (Table 2).

Do Cancer, CVD, and ND Explain the Association of the e4 Allele with Lifespan?

To address this question, we focused on the more homogeneous groups of participants of the FHS original and FHSO cohorts defined in the previous subsection (the LLFS sample is underpowered for such analyses) in order to diminish bias attributable to disproportionality of hazards when using the Cox regression model. Given slightly smaller samples of the FHS participants with known ND status (Table 1), these analyses were limited to individuals with missing information on ND excluded (sample sizes are provided in the respective tables along with the effect estimates).

Additive adjustments of the Cox regression models estimating the risk of death for carriers and non-carriers of the e4 allele by (i)

CVD, (ii) CVD and cancer, and (iii) CVD, cancer, and ND, reveal that CVD and cancer do not explain the observed associations. Contrarily, CVD and cancer tend to improve the estimates in each sample with a more pronounced role for cancer (Figure 3). ND plays at most minor mediating role in the associations of the e4 allele with survival of either men (Figure 3A) or women (Figure 3B). Thus, none of these diseases explain the association of the e4 allele with risks of death (see Supplementary Information, Table S1).

Do Cancer, CVD, and ND Nonlinearly Modulate the Effect of the e4 Allele on Lifespan?

Given no qualitative difference in the additive role of CVD, cancer, and ND in the e4-specific risks of death across the FHS samples, we evaluated the risks in the largest more homogeneous pooled sample of the FHS and FHSO participants in disease-stratified analyses (see Methods). Figure 4 and Table 3 show that the risks of death for women are the same regardless of CVD or ND status, i.e., neither CVD nor ND increase mortality of the e4 female carriers nonlinearly even after adjustment for alternative



Figure 2. Empirical age patterns of survival of the ApoE4 carriers and non-carriers in the LLFS. Patterns are shown for (A and C) men, (B and D) women; and for (A and B) long-living individuals (LLFS_P) and (C and D) offspring of long-living individuals and spouses (LLFS_O+S) who carry (E4) and do not carry (NoE4) the ApoE4 allele. By design, the LLFS included long-living individuals who were aged 80+ years at entry. Offspring of long-living individuals and spouses were pooled together because of the small number of deaths among them (Table 1). The numbers in the insets show the total number of genotyped individuals and the number of deaths among them. doi:10.1371/journal.pgen.1004141.g002

diseases. These diseases do not non-additively modulate men's survival either.

A striking result was that non-skin cancer significantly (p = 0.029) for multiplicative interaction of cancer with ApoE) differentiated the e4-specific risks of death for women from the more homogeneous group (with moderate lifespans of 70 to 95 years) increasing them by 52% from RR = 1.36 (p = 3.8×10^{-1}) for women who did not have cancer to RR = 2.07 (p = 5.0×10^{-7}) for women who had cancer (Figure 4B and Table 3). The high risk of death for women with moderate lifespan who had cancer explained the 3.2-year shorter life expectancy for the e4-allele carriers compared to the non-carriers (Table 4). The same trend on the e4-specific excess in the risks of death was seen for male cancer patients compared to non-patients (Figure 4A). Cancer increases risks for the e4 allele carriers compared to the noncarriers making them marginally significant, RR = 1.31 (p = 0.080) (Table 3).

The available sample size allowed us to gain some insights on potential differences between cancer sites (other than skin) in these associations. In these analyses we excluded major sex-specific sites, i.e., prostate in men and breast in women. Figure 5 and Table 3 show that relative risks of death for men without non-sex-specific cancers (RR = 1.11) increases compared to men without cancers (RR = 1.03) but it declines for men having non-sex-specific cancers (RR = 1.17) compared to men having cancers (RR = 1.31). This pattern suggests that the potential modulating effect of cancer in men is likely not sensitive to cancer site. Contrary to men, Figure 5 and Table 3 show that modulating role of cancer in women is entirely attributed to non-sex-specific cancers. The relative risk of death for women with moderate lifespan who had non-sex-specific cancers became much more pronounced (RR = 2.51), $p = 5.3 \times 10^{-8}$). This high risk explained the 4.2-year difference in life expectancy for the e4-allele carriers and non-carriers in this group (Table 4).

Table 2. Relative risks of death for the ApoE4 allele carriers compared to the non-carriers in the selected age groups of the genotyped participants of the FHS original, FHSO, and LLFS cohorts.

Cohort	Age group	N _{total}	N _{died}	RR	р	95% CI
Men						
FHS	All	456	407	1.16	0.239	0.91-1.48
FHSO	All	1874	457	1.08	0.492	0.87-1.33
FHS+FHSO	All	2330	864	1.12	0.178	0.95-1.31
LLFS_P*	All	661	322	1.52	6.9×10 ⁻³	1.12-2.06
LLFS_P	All	661	322	1.17	0.319	0.86–1.60
LLFS_O+S	All	1430	58	0.81	0.537	0.41-1.59
FHS	≥95	29	22	2.00	0.214	0.67-5.96
FHS	<95	427	385	1.18	0.195	0.92-1.53
FHSO	≥70	892	277	1.13	0.365	0.87-1.48
FHSO	<70	982	180	0.97	0.831	0.71-1.32
FHS+FHSO	≥70-<95	1319	662	1.17	0.096	0.97-1.40
LLFS_O+S	≥70	484	44	0.64	0.294	0.28–1.47
LLFS_O+S	<70	946	14	1.20	0.328	0.37-3.86
Women						
FHS	All	802	649	1.25	2.7×10^{-2}	1.03–1.52
FHSO	All	2050	284	1.59	2.4×10^{-4}	1.24-2.05
FHS+FHSO	All	2852	933	1.36	1.3×10^{-4}	1.16–1.60
LLFS_P*	All	722	317	0.98	0.924	0.66-1.45
LLFS_P	All	722	317	0.78	0.279	0.50-1.22
LLFS_O+S	All	1845	63	2.23	5.2×10 ⁻³	1.27-3.90
FHS	≥95	126	90	0.94	0.794	0.57–1.55
FHS	<95	676	559	1.37	1.7×10 ⁻³	1.12–1.66
FHSO	≥70	987	188	1.80	1.3×10^{-4}	1.33–2.43
FHSO	<70	1063	96	1.10	0.638	0.74–1.65
FHS+FHSO	≥70-<95	1663	747	1.48	3.6×10^{-6}	1.26–1.75
LLFS_O+S	≥70	596	50	3.04	7.8×10^{-4}	1.59–5.81
LLFS_O+S	<70	1249	13	0.40	0.394	0.05-3.28

RR = relative risk; CI = Confidence interval; N_{total} and N_{died} denote the total number of genotyped individuals and the number of deaths among them, respectively.

All models were adjusted for birth cohorts measured as a continuous variable except the model for long-living men and women from the LLFS parental generation (LLFS_P) denoted by asterisk (^{*}). Models for pooled samples in the FHS (i.e., the FHS original and FHSO cohorts; denoted as FHS+FHSO) and LLFS (i.e., the LLFS offspring and spouses; denoted as LLFS_O+S) were adjusted for potential cohort differences. Models for the LLFS were also adjusted for potential field center differences.

"All" in column "Age group" denotes the sample of all ages; other notations in this column indicate the range of ages at death or the end of follow up in each sample. For example, " \geq 70" implies a group of individuals who died at 70+ years or was aged 70+ years at the end of follow up.

doi:10.1371/journal.pgen.1004141.t002

Discussion

The e4 Allele and Human Lifespan

Analysis of genotyped offspring in the FHS revealed that the e4 allele is irrelevant to survival in mid to early-old life, up to about 70 years (Table 2). This result appeared to be corroborated in an independent population of the LLFS offspring and spouses (Table 2). The e4 allele changed its role from neutral in mid to early-old life to detrimental at older ages. This change was found

in independent samples of the FHS Offspring cohort (Figure 1D) and the LLFS offspring and spouses (Figure 2D). Moreover, this change occurred concordantly in the FHSO and LLFS: (i) at about the same age of 70 years and (ii) in women only. The detrimental effect of the e4 allele at old ages (until 95 years of age) was also found in a sample of the FHS women (Figure 1B; note that virtually no individuals with lifespan less than 70 years were genotyped in this cohort).

At extreme ages (95 years and older) we concordantly observed a neutral role of the e4 allele in each gender in the FHS (Figures 1A and 1B). Analysis of the long-living individuals in the LLFS corroborated these findings (see the "**Empirical Age Patterns** of Survival of the FHS and LLFS Men and Women" and "**Risks of Death of the FHS and LLFS Men and Women**" subsections).

Overall, these analyses demonstrated a strong detrimental effect of the e4 allele on survival which was mostly attributed to women with moderate lifespans of 70 to 95 years in the FHS, FHSO, and LLFS. For example, the e4 allele increased the risks of death of the FHS and FHSO women by about 48% (RR = 1.48) with very high confidence, $p = 3.6 \times 10^{-6}$ (Table 2).

Although our study provided robust evidence of a womenspecific detrimental effect of the e4 allele on lifespan in three different samples of mostly North-American population (i.e., FHS, FHSO, and LLFS, see Methods), there is also robust evidence of a detrimental effect of this allele in Swedish men but not women [17]. Further, although our results on the neutral role of the e4 allele at extreme ages (95 years and older) are in agreement with some meta-analyses [e.g., 32], there is also evidence of a significant detrimental effect of the e4 allele at those ages in the Danish population [18]. The results by Rosvall et al. [17], Jacobsen et al. [18], and ours explicitly show that the effect of the e4 allele on lifespan may not be the same in different populations. These robust evidences from different populations illustrate that the concept of replication of the same effect of the same allele on the same complex phenotype characteristic for post-reproductive period has inherent limitations [33-36].

The e4 Allele, Human Lifespan, and Additive Effects of Major Diseases

The e4 allele is a major susceptibility allele for Alzheimer disease (which is a subtype of the ND in this study) particularly in Caucasians [4] (but may be not in Hispanics [37]). Despite that, our well-powered analyses show that ND explains at most a tiny part in the association of the e4 allele with survival (Figure 3). The results of our analyses do not support the hypothesis that the lack of a mediating effect of ND can be due to potential ND misclassification. This is evidenced in Figure 3 by: (i) the tiny reduction of the effect size attributed to ND (Figure 3) despite the large prevalence of ND (particularly in the FHS as the older cohort, Table 1), and (ii) the role of cancer as a nonlinear modulator of the effect of the e4 allele and dementia to survival was also observed in other studies [16] although the attenuation of the effect size by dementia varied [17].

Despite the associations of the e4 allele with CVD [6,7] and with CVD-free life [19,38], our analyses show that CVD does not explain the effect of the e4 allele on women's survival (Figure 3). Recent analyses support these results by showing independent associations of the e4 allele and various characteristics of cardiovascular health and CVD with survival [16,17,39].

Several studies reported on a role of the ApoE gene in cancer [40–43]. It has been also shown that the e4 allele can increase cancer-free lifespan in the FHS and FHSO men [19,38]. The



Figure 3. Relative risks of death and p-values for the ApoE4 allele carriers compared to the non-carriers. The risks were evaluated in more homogeneous groups of (A) men and (B) women who died or were right censored at ages: (i) younger than 95 years in the FHS, (ii) 70 years and older in the FHSO, and (iii) 70 to 95 years in the pooled sample of the FHS and FHSO cohorts. The basic model denotes adjustment for birth cohorts (all models) and an indicator of the FHS or FHSO in the pooled sample (FHS+FHSO). Adjustments by diseases are additional to the basic adjustment. Thin bars show 95% confidence intervals. Exact numeric values for the estimates and sample sizes are given in Supplementary Table S1. Right y-axes show (A) minus log-base-10-transformed p-values. (A) The horizontal line and (B) upper x-axis show the conventional level of significance, i.e., $|log_{10}(0.05)| = 1.3$. doi:10.1371/journal.pgen.1004141.g003

analyses in this study show no mediating role of cancer in the association of the e4 allele with women's survival; the additive contribution of cancer, however, can modulate the effect of the e4 allele, increasing the strength of this association (Figure 3).

CVD and cancer are the most common causes of death in humans and ND is fast growing cause of death in the elderly. CVD and ND are the diseases which have been most consistently associated with ApoE4. Despite that, these diseases do not explain



Figure 4. Disease-stratified relative risks of death and p-values for the ApoE4 allele carriers compared to the non-carriers. The risks were evaluated in a more homogeneous group of (A) men and (B) women who died or were right censored at ages 70 to 95 years in the pooled sample of the FHS and FHSO cohorts. The models were adjusted for birth cohorts, an indicator of the FHS or FHSO, and additive contributions of CVD, ND, and cancer, as applicable, e.g., the model for samples stratified by CVD was adjusted by cancer and ND. Multiplicative interaction between ApoE and cancer for women is significant (p = 0.029). Thin bars show 95% confidence intervals. Exact numeric values for the estimates and sample size are given in Table 3. Right y-axes show (A) minus log-base-10-transformed p-values and (B) log-base-10-transformed p-values. (A) The horizontal line and (B) upper x-axis show the conventional level of significance, i.e., $|log_{10}(0.05)| = 1.3$.

Table 3. Disease-stratified relative risks of death for the ApoE4 carriers compared to the non-carriers in the more homogeneous group of the FHS and FHSO participants with lifespans of 70 to 95 years.

Disease	Disease status	N _{total}	N _{died}	RR	p	95% CI
Men [*]						
CVD	No	614	208	1.18	0.262	0.88-1.59
	Yes	675	442	1.14	0.313	0.89–1.45
Cancer	No	810	355	1.03	0.810	0.80-1.33
	Yes	479	295	1.31	0.080	0.97–1.78
Non-prostate	No	981	431	1.11	0.390	0.88-1.40
	Yes	308	219	1.17	0.419	0.80-1.70
ND	No	1159	527	1.10	0.376	0.89–1.36
	Yes	130	123	1.22	0.431	0.75–1.98
Women [*]						
CVD	No	955	285	1.64	1.4×10^{-4}	1.27–2.11
	Yes	680	447	1.51	2.8×10^{-4}	1.21–1.88
Cancer	No	1203	475	1.36	3.8×10 ⁻³	1.11–1.68
	Yes	432	257	2.07	5.0×10 ⁻⁷	1.56–2.75
Non-breast	No	1360	552	1.36	1.7×10 ⁻³	1.12–1.66
	Yes	275	180	2.51	5.3×10^{-8}	1.80-3.49
ND	No	1435	543	1.63	1.8×10 ⁻⁶	1.33–1.99
	Yes	200	189	1.49	1.4×10 ⁻²	1.08-2.04

*Individuals with missing neurodegenerative disorders (ND) status were excluded in all models.

CVD = cardiovascular diseases; Cancer includes all sites but skin; Non-prostate indicates non-skin cancers apart from prostate neoplasm in men; Non-breast indicates non-skin cancers apart from breast neoplasm in women;

RR = relative risk; CI = Confidence interval; N_{total} and N_{died} denote the total number of genotyped individuals and the number of deaths among them, respectively.

All models were adjusted for birth cohorts, an indicator of the FHS or FHSO, and additive contribution of CVD, ND, and cancer, as applicable, e.g., the model for samples stratified by CVD was adjusted by cancer and ND. doi:10.1371/journal.pgen.1004141.t003

the detrimental role of the e4 allele in lifespan. This finding implies the existence of a mechanism linking the e4 allele with lifespan which is largely independent of the mechanisms affecting susceptibility to CVD, cancer, and ND. Given also that the e4 allele may not be associated with frailty [14,44], it is likely that this allele can be directly involved in regulation of human aging through intrinsic biological mechanisms. One potential mechanism could be associated with inflammation which may be involved in aging through two main pathways associated with "immunosenescence and synergies with chronic diseases that have inflammatory components" [29]. Given no mediating role of CVD, cancer, and ND observed in our study and that these diseases (particularly CVD and ND) can have e4-specific inflammatory etiology [8,45], it might well be the case that the e4 allele affects survival through immunosenescence whereas it affects the risks of diseases through disease-specific inflammatory component.

The e4 Allele, Human Lifespan, and Non-Additive Effects of Maior Diseases

Neither CVD nor ND non-additively (i.e., nonlinearly) modulates the detrimental effect of the e4 allele on women's survival, i.e., the relative risks of death for the e4 allele carriers are the same regardless of women's CVD and ND statuses (Figure 4B). This result is in line with findings by Little et al. [16]. However, the e4 allele was shown to be mostly associated with dementia-caused deaths by Newman et al. [39].

We found that cancer showed a significant nonlinear modulating effect in the association of the e4 allele with women's survival (Figure 4B). The e4-positive female cancer patients have about a two-fold increased risk of death at ages between 70 and 95 years compared to the non-e4 allele carriers (RR = 2.07) which is highly significant, $p = 5.0 \times 10^{-7}$ (Table 3). Such a strong effect results in a 3.2-year shorter life expectancy of the e4 carriers compared to the non-carriers in this sample (Table 4). Further analyses show that this effect is attributed to non-sex-specific cancer sites, it substantially increases, i.e., RR = 2.51, $p = 5.3 \times 10^{-8}$ (Table 3), and it explains the large 4.2 year differential in the life expectancy (Table 4). Women without cancer carrying the e4 allele are still at significant risk of death. The same non-additive role of cancer was found in the effect of the e4 allele on men's survival, i.e., this allele negatively affected cancer survivorship (Figure 4A). The diminished role of cancer as a nonlinear modulator of the effect of the e4 allele on survival in men compared to women can be attributed to a protective role of this allele in susceptibility to risk of cancer in men but not in women [19,38], i.e., protection against risks of cancer may well explain modest risks of cancer survivorship of male e4 carriers.

Table 4. Kaplan-Meier estimates of life expectancy of the FHS and FHSO women from the more homogeneous group who were aged between 70 and 95 years at death or the end of follow up in 2008 stratified by cancer and the ApoE4 statuses.

Cancer type	Cancer status	E4 allele	N _{total}	N _{died}	LE, years	95% CI
All sites but skin	no	no	926	353	88.3	87.8-88.8
		yes	277	122	87.1	86.3-88.0
	yes	no	346	195	86.6	85.9–87.4
		yes	86	62	83.4	82.0-84.7
All sites but skin & breast	no	no	1052	412	88.2	87.7–88.6
		yes	308	140	87.1	86.2-87.9
	yes	no	220	136	86.2	85.2-87.1
		yes	55	44	82.0	80.4-83.5

LE = life expectancy; CI = confidence interval; N_{total} and N_{died} denote the total number of genotyped individuals and the number of deaths among them, respectively. doi:10.1371/journal.pgen.1004141.t004



Figure 5. Cancer-stratified relative risks of death and log-base-10-transformed p-values for the ApoE4 allele carriers compared to the non-carriers. The risks were evaluated in more homogeneous groups of individuals who died or were right censored at ages 70 to 95 years in the pooled sample of the FHS and FHSO. "No" indicates individuals who did not have non-skin cancers apart from prostate neoplasm in men or breast neoplasm in women. "Yes" indicates individuals who had non-skin cancers other than prostate neoplasm in men or breast neoplasm in women. The models were adjusted for birth cohorts, an indicator of the FHS or FHSO, and additive contribution of CVD and ND. The multiplicative interaction between ApoE and non-sex-specific cancer in women was highly significant $(p = 5.2 \times 10^{-3})$. Thin bars show 95% confidence intervals. Exact numeric values for the estimates and sample size are given in Table 3 (nonprostate and non-breast). The solid horizontal line shows the conventional level of significance, i.e., $log_{10}(0.05) = -1.3$. doi:10.1371/journal.pgen.1004141.g005

The cancer-sensitive non-additive effect of the e4 allele on human lifespan suggests that mechanisms associated with cancer survivorship (i.e., with its progression and/or treatment) can interfere with a mechanism linking the e4 allele to lifespan. Our findings are particularly in line with inflammatory pathways [29,43] which may overlap for aging and cancer survivorship as a result of the compromising of the immune system with age [46] (see also next subsection). Thus, the non-additive role of cancer in the effect of the e4 allele on lifespan and the lack of this role for CVD and ND likely underscores the synergism between cancer and aging.

The e4 Allele and Survival in Humans

Given the persistence of the e4 allele in humans, it may be beneficial in early life and, thus, be subject to balancing selection [27–30]. Indeed, several studies provided support for a beneficial role of the e4 allele in early life. For example, it was shown that the proportion of the e4 allele was significantly smaller in spontaneously aborted embryos than in adults [47]. The proportion of the e4 allele was also found to be significantly larger in healthy liveborn infants compared with stillborn infants and with adults [48]. These findings suggest that the e4 allele can benefit early survival. Then, given the detrimental role of this allele for survival in old ages, we should expect a neutral role at some point in between. Our finding of a neutral role of the e4 allele in survival in mid to early-old life of the genotyped FHS and LLFS participants supports this logic.

Studies also show that ApoE4 may protect against early life infectious diseases such as, e.g., diarrhea [49] and liver damage caused by the hepatitis C virus infection [50,51]. A putative protective mechanism may be associated with an enhanced function of the immune system in early life [25] with a role of ApoE as an immunomodulator [52]. At old ages immunosenescence may be a factor favoring neoplasia [53]. Then, if ApoE4 boosts the immune system in early life, this may naturally lead to prematurely exhausting this system later in life which may affect cancer survivorship for carriers of this allele (and, thus, implying antagonistic pleiotropy). This hypothesis is supported by our findings of a strong non-additive modulating role of cancer in survival of female e4 allele carriers (Figure 5), by the very high proportion of deaths (80%) among female e4 carriers with nonsex-specific cancer by age 95 years (44 deaths among 55 carriers; Table 4), and by the 150% excess risk of death for such women compared to the non-e4 carriers (RR = 2.51, p = 5.3×10^{-8} ; Table 3). These high death rates can, in part, explain the diminishing detrimental effect of ApoE4 at very advanced ages (95+ years) in the FHS.

The lack of an association of ApoE4 with survival at extreme ages (95+) in the FHS and in an exceptional population of the LLFS long-living participants suggests that the detrimental effect of ApoE4 can be counterbalanced in some individuals. Potential factors can include buffering mechanisms (by other genes [54]) and/or environmental modulations of genetic effects [36]. Given large samples of long-living individuals in the LLFS, this study could be highly promising for revealing such mechanisms.

Concluding Remarks

Analyses of the association of the ApoE4 allele with lifespan in three populations of the FHS, FHSO, and LLFS participants showed that women's lifespan was more sensitive to the e4 allele than men's. The adverse role of the e4 allele was limited to women with moderate lifespans of about 70 to 95 years; no survival disadvantage is seen for women with lifespans less than 70 or more than 95 years. The highly significant association of the e4 allele with lifespan was not explained by major diseases including CVD, ND, and cancer, whose risks can be sensitive to this allele, in large FHS samples. Non-skin cancer non-additively increased mortality of the FHS women with moderate lifespans increasing the risks of death of the e4 carriers two-fold compared to the non-carriers. High and highly significant risks of death of the e4-allele carriers in this sample explained their 3.2 year shorter life expectancy. The results suggest a pivotal role of non-sex-specific cancer as a nonlinear modulator of survival in this sample of women that increased the risk of death of the ApoE4 carriers by 150% $(p = 5.3 \times 10^{-8})$ compared to the non-carriers and explained the 4.2 year differential in life expectancy in this group. Our results suggest the existence of age- and gender-sensitive systemic mechanisms linking the e4 allele to lifespan which can nonadditively interfere with cancer-related mechanisms.

Methods

Data

The Framingham Heart Study (FHS). The original (FHS) cohort was launched in 1948 in Framingham, Massachusetts. 22 years later a cohort of offspring of participants of the FHS original cohort was launched (known as the FHS Offspring or FHSO cohort). The study design has been previously described [55–57]. Briefly, the FHS includes N = 5,209 respondents aged 28–62 years at baseline who have been biennially followed during nearly 60 years. The FHSO respondents (N = 5,124) aged 5–70 years at baseline were biological descendants (N = 3,514), their spouses (N = 1,576), and adopted offspring (N = 34) of the FHS

The FHS/FHSO participants have been followed for the onset of CVD, cancer, and death through regular examinations at the FHS clinic, surveillance of hospital admissions, and death registries since baseline examinations in the FHS and FHSO [55,56], currently through 2008. Dementia-free survivors who attended examination 14 in the FHS and examination two in the FHSO were continuously followed for onset of dementia and Alzheimer disease [58], currently through 2008.

Biospecimens were mostly collected in the late 1980s, and through 1990s, from surviving participants [59,60]. The procedure used for the ApoE genotyping was described in Lahoz et al. [59]. The data available for this study include information on the ApoE2/3/4 polymorphism for the 1,258 FHS and 3,924 FHSO participants.

The Long Life Family Study (LLFS). The LLFS collected data in about equal proportions at four field centers (three in the U.S., i.e., Boston, New York, and Pittsburg, and one in Denmark) on families showing exceptional familial longevity (virtually all participants were whites). The study eligibility criteria were described elsewhere [61–63].

Briefly, in the U.S., the families eligible for the LLFS must have two living siblings aged 80+ years, two living offspring of one or more of the siblings, and a living spouse of one of the offspring who were considered as controls. In addition, the family must demonstrate exceptional longevity based on a Family Longevity Selection Score, which is a summary-measure based on the survival experience of the oldest living generation of siblings relative to what would be expected based on birth cohort life tables [61]. Families with members of this generation who were still alive and larger sibships were given higher priorities. Finally, an eligible family was enrolled in the LLFS if at least 3 family members (the proband, at least one sibling of the proband, and one offspring of the proband or the sibling) indicated their willingness to participate.

In Denmark, individuals who would be aged 90+ years during the study recruitment period were first identified in the Danish National Register of Persons [62]. Then, using information on the place of birth and the names, parish registers available in regional archives were searched to locate the parents of the elderly individuals in order to identify sibships. The identified subjects were contacted to further assess the family's eligibility for participation in the LLFS using criteria parallel to that used in the U.S.

Information from the 4,954 U.S. and Danish LLFS participants was collected using similar questionnaires and in-home physical examinations at baseline between 2006 and 2009. Once enrolled, the LLFS participants were followed longitudinally. During the follow up for about 6 years (currently through April 2013) selfreported information on diseases collected at baseline was updated and information on vital status was collected.

Biospecimens were collected at baseline. Genotyping of the ApoE polymorphism was conducted using procedures detailed elsewhere [31]. The data include information on the ApoE2/3/4 polymorphism for the 4,659 LLFS participants including long-living individuals (N = 1,384, probands and siblings), their offspring (N = 2,321, children, nieces, and nephews), spouses of long-living individuals (N = 177), and spouses of offspring of long-living individuals (N = 777). Due to small numbers of spouses of

References

the long-living individuals, they were pooled together with spouses of offspring (N = 954).

Analysis

We use data on longitudinally followed FHS/FHSO and LLFS participants to characterize the role of the ApoE4 allele (e2/4, e3/4, and e4/4) and non-e4 genotypes (e2/2, e2/3, and e3/3) in the lifespans of men and women separately.

Associations of the e4 allele with risks of death were characterized by the Kaplan-Meier estimator and the Cox proportional hazards regression model. The time variable in the analyses was age at death or age at the end of follow up. The model adjustments were explicitly stated when applicable.

To examine whether or not major human diseases can shape the association of the e4 allele with survival, we considered additive and nonlinear roles of CVD (diseases of hearth and stroke combined), cancer, and ND (dementia and Alzheimer disease combined) in this association. We considered all non-skin cancers unless explicitly stated. CVD and ND were chosen because they were most consistently associated with the ApoE polymorphism [3,6,7,64]. Recent studies also showed that the ApoE polymorphism can be associated with cancer [e.g., 41]. These analyses were conducted using rigorously ascertained information on diseases in the FHS/FHSO only because the LLFS data are currently underpowered for such analyses.

To address nonlinear effect of diseases on the association of the e4 allele with survival, we conducted disease-stratified analyses. Each disease group included individuals who were diagnosed with the disease (or died from it) prior to death or the end of follow up in 2008. Otherwise, individuals were included in the complementary non-disease group [65].

We used the robust sandwich estimator of variances in the Cox model to account for potential clustering (e.g., familial). Statistical analyses were conducted using SAS (release 9.3, Cary, NC, USA).

This study used de-identified data from the FHS and LLFS. The FHS data are available from the NHLBI through dbGaP. No new data were collected in this work. As such, this study does not require either ethics committee approval or written consent.

Supporting Information

Table S1 Disease-conditional and unconditional relative risks of death for the ApoE4 allele carriers compared to the non-carriers in the selected age groups of the genotyped participants of the FHS original and FHSO cohorts.

(\mathbf{DOC})

Acknowledgments

We thank four reviewers for their valuable and constructive comments which have improved this paper.

Author Contributions

Conceived and designed the experiments: AMK IC. Performed the experiments: AMK KGA LA. Analyzed the data: AMK. Contributed reagents/materials/analysis tools: NS. Wrote the paper: AMK. Interpretation of the results: IC. Assessing the logic of the final version: SVU ES AIY. Discussion of the results: AMK IC SVU ES AIY KC NS MAP. Discussion of phenotypes: AMK KC NS.

 McIntosh AM, Bennett C, Dickson D, Anestis SF, Watts DP, et al. (2012) The Apolipoprotein E (APOE) Gene Appears Functionally Monomorphic in Chimpanzees (Pan troglodytes). PLoS One 7: e47760.

Fullerton SM, Clark AG, Weiss KM, Nickerson DA, Taylor SL, et al. (2000) Apolipoprotein E variation at the sequence haplotype level: implications for the origin and maintenance of a major human polymorphism. Am J Hum Genet 67: 881–900.

- Corder EH, Saunders AM, Strittmatter WJ, Schmechel DE, Gaskell PC, et al. (1993) Gene dose of apolipoprotein E type 4 allele and the risk of Alzheimer's disease in late onset families. Science 261: 921–923.
- Bertram L, McQueen MB, Mullin K, Blacker D, Tanzi RE (2007) Systematic meta-analyses of Alzheimer disease genetic association studies: the AlzGene database. Nat Genet 39: 17–23.
- Farrer LA, Cupples LA, Haines JL, Hyman B, Kukull WA, et al. (1997) Effects of age, sex, and ethnicity on the association between apolipoprotein E genotype and Alzheimer disease. A meta-analysis. APOE and Alzheimer Disease Meta Analysis Consortium. JAMA 278: 1349–1356.
- Song Y, Stampfer MJ, Liu S (2004) Meta-analysis: apolipoprotein E genotypes and risk for coronary heart disease. Ann Intern Med 141: 137–147.
- Wilson PW, Schaefer EJ, Larson MG, Ordovas JM (1996) Apolipoprotein E alleles and risk of coronary disease. A meta-analysis. Arterioscler Thromb Vasc Biol 16: 1250–1255.
- Kulminski AM, Culminskaya I, Arbeev KG, Ukraintseva SV, Stallard E, et al. (2013) The role of lipid-related genes, aging-related processes, and environment in healthspan. Aging Cell 12: 237–246.
- Soerensen M, Dato S, Tan Q, Thinggaard M, Kleindorp R, et al. (2013) Evidence from case-control and longitudinal studies supports associations of genetic variation in APOE, CETP, and IL6 with human longevity. Age (Dordr) 35: 487–500.
- Christensen K, Johnson TE, Vaupel JW (2006) The quest for genetic determinants of human longevity: challenges and insights. Nat Rev Genet 7: 436–448.
- Napolioni V, Gianni P, Carpi FM, Predazzi IM, Lucarini N (2011) APOE haplotypes are associated with human longevity in a Central Italy population: evidence for epistasis with HP 1/2 polymorphism. Clin Chim Acta 412: 1821– 1824.
- Deelen J, Beekman M, Uh HW, Helmer Q, Kuningas M, et al. (2011) Genomewide association study identifies a single major locus contributing to survival into old age; the APOE locus revisited. Aging Cell 10: 686–698.
- Nebel A, Kleindorp R, Caliebe A, Nothnagel M, Blanche H, et al. (2011) A genome-wide association study confirms APOE as the major gene influencing survival in long-lived individuals. Mechanisms of Ageing and Development 132: 324–330.
- Jazwinski SM, Kim S, Dai J, Li L, Bi X, et al. (2010) HRAS1 and LASS1 with APOE are associated with human longevity and healthy aging. Aging Cell 9: 698–708.
- Ewbank DC (2004) The APOE gene and differences in life expectancy in Europe. J Gerontol A Biol Sci Med Sci 59: 16–20.
- Little DM, Crooks VC, Petitti DB, Chiu V, Schellenberg GD, et al. (2009) Mortality, dementia, and apolipoprotein E genotype in elderly white women in the United States. J Am Geriatr Soc 57: 231–236.
- Rosvall L, Rizzuto D, Wang HX, Winblad B, Graff C, et al. (2009) APOErelated mortality: effect of dementia, cardiovascular disease and gender. Neurobiol Aging 30: 1545–1551.
- Jacobsen R, Martinussen T, Christiansen L, Jeune B, Andersen-Ranberg K, et al. (2010) Increased effect of the ApoE gene on survival at advanced age in healthy and long-lived Danes: two nationwide cohort studies. Aging Cell 9: 1004–1009.
- Kulminski AM, Culminskaya I, Ukraintseva SV, Arbeev KG, Arbeeva L, et al. (2011) Trade-off in the effects of the apolipoprotein E polymorphism on the ages at onset of CVD and cancer influences human lifespan. Aging Cell 10: 533–541.
- Schwanke CH, da Cruz IB, Leal NF, Scheibe R, Moriguchi Y, et al. (2002) Analysis of the association between apolipoprotein E polymorphism and cardiovascular risk factors in an elderly population with longevity. Arq Bras Cardiol 78: 561–579.
- Ruiu G, Cassader M, Gambino R, Alemanno N, Demichieli F, et al. (1995) Apolipoprotein E allele frequencies in an Italian population: relation to age and lipid profile. Aging (Milano) 7: 185–189.
- Galinsky D, Tysoe C, Brayne CE, Easton DF, Huppert FA, et al. (1997) Analysis of the apo E/apo C-I, angiotensin converting enzyme and methylenetetrahydrofolate reductase genes as candidates affecting human longevity. Atherosclerosis 129: 177–183.
- Panza F, Frisardi V, Capurso C, D'Introno A, Colacicco AM, et al. (2009) Apolipoprotein E, dementia, and human longevity. J Am Geriatr Soc 57: 740– 742.
- Lewis SJ, Brunner EJ (2004) Methodological problems in genetic association studies of longevity-the apolipoprotein E gene as an example. Int J Epidemiol 33: 962–970.
- Finch CE (2012) Evolution of the human lifespan, past, present, and future: phases in the evolution of human life expectancy in relation to the inflammatory load. Proc Am Philos Soc 156: 9–44.
- Drenos F, Kirkwood TB (2010) Selection on alleles affecting human longevity and late-life disease: the example of apolipoprotein E. PLoS One 5: e10022.
- 27. Charlesworth B (1996) Evolution of senescence: Alzheimer's disease and evolution. Curr Biol 6: 20–22.
- Martin GM (1999) APOE alleles and lipophylic pathogens. Neurobiol Aging 20: 441–443.
- Finch CE (2010) Evolution in health and medicine Sackler colloquium: Evolution of the human lifespan and diseases of aging: roles of infection, inflammation, and nutrition. Proc Natl Acad Sci U S A 107 Suppl 1: 1718–1724.

- 30. Kuhlmann I, Minihane AM, Huebbe P, Nebel A, Rimbach G (2010) Apolipoprotein E genotype and hepatitis C, HIV and herpes simplex disease risk: a literature review. Lipids Health Dis 9: 8.
- Schupf N, Barral S, Perls T, Newman A, Christensen K, et al. (2013) Apolipoprotein E and familial longevity. Neurobiol Aging 34: 1287–1291.
- Ewbank DC (2002) Mortality differences by APOE genotype estimated from demographic synthesis. Genet Epidemiol 22: 146–155.
- Yashin AI, Wu D, Arbeev KG, Ukraintseva SV (2012) Polygenic Effects of Common Single-Nucleotide Polymorphisms on Life Span: When Association Meets Causality. Rejuvenation Res 15: 381–94.
- Day-Williams AG, Zeggini E (2011) The effect of next-generation sequencing technology on complex trait research. Eur J Clin Invest 41: 561–567.
- McClellan J, King MC (2010) Genetic heterogeneity in human disease. Cell 141: 210–217.
- Kulminski AM (2013) Unraveling genetic origin of aging-related traits: evolving concepts. Rejuvenation Res 16: 304–312.
- Campos M, Edland SD, Peavy GM (2013) Exploratory study of apolipoprotein E epsilon4 genotype and risk of Alzheimer's disease in Mexican Hispanics. J Am Geriatr Soc 61: 1038–1040.
- Kulminski AM, Culminskaya I, Arbeev KG, Ukraintseva SV, Arbeeva L, et al. (2013) Trade-off in the effect of the APOE gene on the ages at onset of cardiocascular disease and cancer across ages, gender, and human generations. Rejuvenation Res 16: 28–34.
- Newman AB, Sachs MC, Arnold AM, Fried LP, Kronmal R, et al. (2009) Total and cause-specific mortality in the cardiovascular health study. J Gerontol A Biol Sci Med Sci 64: 1251–1261.
- Saadat M (2012) Apolipoprotein E (APOE) Polymorphisms and Susceptibility to Breast Cancer: A Meta-Analysis. Cancer Res Treat 44: 121–126.
- Chen YC, Pohl G, Wang TL, Morin PJ, Risberg B, et al. (2005) Apolipoprotein E is required for cell proliferation and survival in ovarian cancer. Cancer Res 65: 331–337.
- Sakashita K, Tanaka F, Zhang X, Mimori K, Kamohara Y, et al. (2008) Clinical significance of ApoE expression in human gastric cancer. Oncol Rep 20: 1313– 1319.
- Slattery ML, Sweeney C, Murtaugh M, Ma KN, Potter JD, et al. (2005) Associations between apoE genotype and colon and rectal cancer. Carcinogenesis 26: 1422–1429.
- Rockwood K, Nassar B, Mitnitski A (2008) Apolipoprotein E-polymorphism, frailty and mortality in older adults. J Cell Mol Med 12: 2754–2761.
- Jofre-Monseny L, Minihane AM, Rimbach G (2008) Impact of apoE genotype on oxidative stress, inflammation and disease risk. Mol Nutr Food Res 52: 131– 145.
- de Magalhaes JP (2013) How ageing processes influence cancer. Nat Rev Cancer 13: 357–365.
- Zetterberg H, Palmer M, Ricksten A, Poirier J, Palmqvist L, et al. (2002) Influence of the apolipoprotein E epsilon4 allele on human embryonic development. Neurosci Lett 324: 189–192.
- Becher JC, Keeling JW, McIntosh N, Wyatt B, Bell J (2006) The distribution of apolipoprotein E alleles in Scottish perinatal deaths. J Med Genet 43: 414–418.
- Oria RB, Patrick PD, Blackman JA, Lima AA, Guerrant RL (2007) Role of apolipoprotein E4 in protecting children against early childhood diarrhea outcomes and implications for later development. Med Hypotheses 68: 1099– 1107.
- Wozniak MA, Itzhaki RF, Faragher EB, James MW, Ryder SD, et al. (2002) Apolipoprotein E-epsilon 4 protects against severe liver disease caused by hepatitis C virus. Hepatology 36: 456–463.
- Fabris C, Toniutto P, Bitetto D, Minisini R, Smirne C, et al. (2005) Low fibrosis progression of recurrent hepatitis C in apolipoprotein E epsilon4 carriers: relationship with the blood lipid profile. Liver Int 25: 1128–1135.
- Vitek MP, Brown CM, Colton CA (2009) APOE genotype-specific differences in the innate immune response. Neurobiol Aging 30: 1350–1360.
- Malaguarnera M, Cristaldi E, Romano G, Malaguarnera L (2012) Autoimmunity in the elderly: Implications for cancer. J Cancer Res Ther 8: 520–527.
- Bergman A, Atzmon G, Ye K, MacCarthy T, Barzilai N (2007) Buffering mechanisms in aging: a systems approach toward uncovering the genetic component of aging. PLoS Comput Biol 3: e170.
- Govindaraju DR, Cupples LA, Kannel WB, O'Donnell CJ, Atwood LD, et al. (2008) Genetics of the Framingham Heart Study population. Adv Genet 62: 33– 65.
- Splansky GL, Corey D, Yang Q, Atwood LD, Cupples LA, et al. (2007) The Third Generation Cohort of the National Heart, Lung, and Blood Institute's Framingham Heart Study: design, recruitment, and initial examination. Am J Epidemiol 165: 1328–1335.
- Cupples LA, Heard-Costa N, Lee M, Atwood LD (2009) Genetics Analysis Workshop 16 Problem 2: the Framingham Heart Study data. BMC Proc 3 Suppl 7: S3.
- Driver JA, Beiser A, Au R, Kreger BE, Splansky GL, et al. (2012) Inverse association between cancer and Alzheimer's disease: results from the Framingham Heart Study. British Medical Journal 344: e1442.
- Lahoz C, Schaefer EJ, Cupples LA, Wilson PW, Levy D, et al. (2001) Apolipoprotein E genotype and cardiovascular disease in the Framingham Heart Study. Atherosclerosis 154: 529–537.

- Myers RH, Schaefer EJ, Wilson PW, D'Agostino R, Ordovas JM, et al. (1996) Apolipoprotein E epsilon4 association with dementia in a population-based study: The Framingham study. Neurology 46: 673–677.
- Sebastiani P, Hadley EC, Province M, Christensen K, Rossi W, et al. (2009) A family longevity selection score: ranking sibships by their longevity, size, and availability for study. Am J Epidemiol 170: 1555–1562.
- Pedersen CB, Gotzsche H, Moller JO, Mortensen PB (2006) The Danish Civil Registration System. A cohort of eight million persons. Dan Med Bull 53: 441– 449.
- Yashin AI, Arbeev KG, Kulminski A, Borecki I, Christensen K, et al. (2010) "Predicting" parental longevity from offspring endophenotypes: data from the Long Life Family Study (LLFS). Mech Ageing Dev 131: 215– 222.
- Eichner JE, Dunn ST, Perveen G, Thompson DM, Stewart KE, et al. (2002) Apolipoprotein E polymorphism and cardiovascular disease: a HuGE review. Am J Epidemiol 155: 487–495.
- Manton KG, Stallard E (1980) Mortality of the chronically impaired. Demography 17: 189–206.